
V13 Sunday Monday Tuesday Wednesday Thursday Friday Saturday

0 14-Jun-09

Tev Beam Off
Cal Calibration
forward forward 

cosmics
switch solenoid 

polarity to 
reverse

reverse forward 
cosmics

Magnet Off Cosmics

Supervised Access 
Survey

Open Pit Door
Stick-Mic Survey

Begin Reduced Shift 
Coverage

L3 cluster machines 
replaced

UPS maintenance
Open Clam Shells

VLPC light leak test

Open EFs
Open CFs

Move master clock 
to free run mode

Open EC
Install South Gap 
Access Hardware
Install North Gap 
Access Hardware
L1Cal PS failure

SMT powered down

Crate 0x18 and 
0x19 excluded

CTT Mixer 12 V PS 
failure

Power down most 
sub-detectors
MCH2 Liebert 
leaking again

Adjust lumi preamp 
ground levels

Feeder 49 
maintenance (9 

hours)

1 21-Jun-09

Majority of 
detector remains 
powered down

Power down AFEs
Site Wide Power 

Outage
Detector Recovery

Collab Meeting

Dismount Lumi 
Monitors

PDT wire extraction
Collab Meeting

D0 Detector Tours
Online production 

DB down
Cal HV 

investigations
Collab Meeting

Cal HV 
investigations

Collab Meeting

Draining LN2 dewar 
39

CAL HV 
Investigations
Collab Meeting

2 28-Jun-09

Reboot DAQ cluster 
for NETAPP 
investigation

HDI recovery effort HDI recovery effort HDI recovery effort HDI recovery effort Lab Holiday

3 5-Jul-09 HDI recovery effort HDI recovery effort HDI recovery effort HDI recovery effort

CAL HV 
Investigations

HDI recovery effort
Purple Haze 
gap bridged

4 12-Jul-09
HDI recovery effort
Crate 0x13 FEBs

HDI recovery effort
Level 2 Re-init tests

SBC tests
HDI recovery effort
Crate 0x13 FEBs HDI recovery effort HDI recovery effort

5 19-Jul-09

HDI recovery effort
Heartbeat SCL 

confused

HDI recovery effort
SMT east side RMI 

trips
Routing Master PS 

failure
HDI recovery effort

Crate 0x72 FEB HDI recovery effort
HDI recovery effort

F Disk Taping

6 26-Jul-09 MCR power off

Power down AFEs
Site Wide Power 

Outage
Safety System Tests
Trigger Framework 

Maintenance
Detector Recovery

HDI recovery effort
Mount North 

Luminosity Monitor

HDI recovery 
efforts

heat exchangers in 
M220 replaced

HDI recovery effort
Mount South 

Luminosity Monitor HDI recovery effort

SMT HV 
investigations
ICD cabled



V13 Sunday Monday Tuesday Wednesday Thursday Friday Saturday

7 2-Aug-09
SMT HV 

investigation SMT HV investigation

SMT HV 
investigations
Online cluster 

kernel upgrades
PPD Noon Tours

SMT HV 
investigation

SMT HV 
investigation

Verify MDT HV 
status

slow control 
power supply 
failure shuts 
down AHU3 
and AHU4

8 9-Aug-09
high humidity in 
the collision hall

SMT HV 
investigation

Test of Collision Hall 
Halon system
Feeder 46b 

maintenance
York chiller back online
SMT HV investigation
Verified lumi monitors 
ready for gap closing

PPD Noon Tours
SMT HV 

investigations
SMT HV 

investigation

Verify gaps ready 
for closing

Detector Checkout

Unrecorded 
Muon and CTT 

Cosmics

9 16-Aug-09

SMT HV 
investigations
Unrecorded 

Muon and CTT 
Cosmics

SMT HV 
investigation 

SMT HV investigations
Unrecorded Muon and 

CTT Cosmics 

Dewar 39 cooldown
PPD Noon Tour

Unrecorded Muon 
and CTT Cosmics

Verify gaps ready 
for closing

Remove south gap 
access hardware
Close south EC

Verify cathedrals 
ready for closing

Remove north gap 
access hardware
Close north EC
Switch to MCR 

clock
Booster Start-Up

Unrecorded 
Muon and CTT 

Cosmics
Booster 

commissioning

10 23-Aug-09

Unrecorded 
Muon and CTT 

Cosmics Housekeeping Day

Databases migrate to 
new machine

DAQ down for IP 
address re-assignment

Close CFs and EFs
Extend bridges and roll 

out nosepieces
Close Clam Shells

Close Pit Access Door
Secure Collision Hall
Test Magnet Power 

Supplies

Magnet power 
supply tests 

continue

D0EVMF switched 
to manual control

Detector Checkout

Production database 
upgrade

Supervised Access 
Resumes

West side scaffold 
installation

Detector Checkout

Clean Service 
Cards on PDTs

Detector Checkout

Magnet off 
cosmics

MI PS start-up

11 30-Aug-09
Magnet off 
cosmics

Resume Full DAQ 
Shift Coverage

Clean Service Cards 
on PDTs

Detector Checkout
Magnet off cosmics

MI  startup
Tevatron cooldown

Dismantle west side 
scaffolding

Detector Checkout
Magnet off cosmics

Stick-Mic Survey

Detector Checkout
Magnet off cosmics

Stick-Mic Survey

Detector Checkout
Magnet off cosmics

Resume Full 
Scheduled Shift 

Coverage
Secure Collision 

Hall
Re-Test Magnet 
Power Supplies

Magnet on cosmics
Magnet on 
cosmics

12 6-Sep-09
Magnet on 
cosmics

Magnet off cosmics
Lab Holiday

Magnet off cosmics
Online network 
isolation test
Tevatron cold Tev PS testing

Establish Tev proton 
beam

Lattice 
measurements

Lifetime 
measurements

13 13-Sep-09
Pbar only store

14x0 store HEP



27-Aug-09



27-Aug-09


